Author's response to reviews

Title: Organizational culture in the Primary Healthcare setting of Cyprus.

Authors:

Theodora Zachariadou (andthe@cytanet.com.cy)
Savvas Zannetos (zannetos@gmail.com)
Andreas Pavlakis (pavlakis@ouc.ac.cy)

Version: 2 Date: 26 September 2012

Author's response to reviews:

Dear Editor,

I am resubmitting the manuscript titled: "Organizational culture in the primary healthcare setting of Cyprus" after considering your comments. I have included the phrase "The study was also approved by the National Bioethics Committee of Cyprus" at page 9 which was not included in the article by mistake.

Yours sincerely

Theodora Zachariadou